

# Central Lancashire Online Knowledge (CLoK)

Title	Women's and communities' views of targeted educational interventions to
	reduce unnecessary caesarean section: a qualitative evidence synthesis
Туре	Article
URL	https://clok.uclan.ac.uk/id/eprint/23655/
DOI	https://doi.org/10.1186/s12978-018-0570-z
Date	2018
Citation	Kingdon, Carol, Downe, Soo and Betran, Ana Pilar (2018) Women's and
	communities' views of targeted educational interventions to reduce
	unnecessary caesarean section: a qualitative evidence synthesis.
	Reproductive Health, 15. p. 130.
Creators	Kingdon, Carol, Downe, Soo and Betran, Ana Pilar

It is advisable to refer to the publisher's version if you intend to cite from the work. https://doi.org/10.1186/s12978-018-0570-z

For information about Research at UCLan please go to <a href="http://www.uclan.ac.uk/research/">http://www.uclan.ac.uk/research/</a>

All outputs in CLoK are protected by Intellectual Property Rights law, including Copyright law. Copyright, IPR and Moral Rights for the works on this site are retained by the individual authors and/or other copyright owners. Terms and conditions for use of this material are defined in the <u>http://clok.uclan.ac.uk/policies/</u>

# REVIEW

**Open Access** 



Women's and communities' views of targeted educational interventions to reduce unnecessary caesarean section: a qualitative evidence synthesis

Carol Kingdon<sup>1\*</sup>, Soo Downe<sup>1</sup> and Ana Pilar Betran<sup>2</sup>

# Abstract

**Background:** There is continued debate about the role of women and communities in influencing rising rates of caesarean section (CS). In settings where CS rates exceed recommended levels, mothers and babies are exposed to potential harms that may outweigh the potential benefits. There is therefore a need to understand how educational interventions targeted at women and communities to reduce unnecessary CS are perceived and used. This qualitative evidence synthesis aimed to explore what women and communities say about the barriers and facilitators to intervention effectiveness for these important groups.

**Method:** Seven electronic databases were searched using predefined search terms. Studies reporting qualitative data pertaining to interventions, published between 1985 and March 2017, with no language restriction were sought. Study quality was independently assessed by two authors before qualitative evidence synthesis was undertaken using an interpretive, meta-ethnography approach. Resulting Statements of Findings were assessed using GRADE-CERQual, and summarised thematically.

**Results:** Twelve studies were included. They were published between 2001 and 2016. Eleven were from high-income countries. Twelve Summaries of Findings encompassed the data, and were graded (moderate or high) on CerQual. The Statements of Findings are reported under three final themes: 1) Mutability of women's and communities' beliefs about birth; 2) Multiplicity of individual information needs about birth; 3) Interactions with health professionals and influence of healthcare system on actual birth method. Women and communities value educational interventions that include opportunities for dialogue, are individualised (including acknowledgement of previous birth experiences), and are consistent with available clinical care and the advice of the health professional they come into contact with.

**Conclusion:** Women's values and preferences for birth, and for information format and content, vary across populations, and evolves in individual women over time. Interactions with health professionals and health system factors can partly be responsible for changes in views. Educational interventions should take into account these dynamic interactions, as well as the women's need for emotional support and dialogue with professionals alongside information about birth. Further research is required to test these findings and the utility of their practical application, particularly in medium and low income settings.

Systematic review registration number: PROSPERO 2017 CRD42017059453.

Keywords: Caesarean section, Educational interventions, Decision-aids, Birth method, Women's views

\* Correspondence: ckingdon@uclan.ac.uk

<sup>1</sup>School of Community Health and Midwifery, Faculty of Health and Wellbeing, University of Central Lancashire, Preston PR1 2HE, UK Full list of author information is available at the end of the article



© The Author(s). 2018 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

# **Plain English summary**

## Why did we do this review?

Caesarean section (CS) can be life-saving in some circumstances. However, current rates suggest that the operation is sometimes used for healthy women and babies when it is not medically necessary. Reasons include health service factors, clinician convenience, and women's choice. If a caesarean section is not medically needed, the benefits can be outweighed by the risks of harm. This is now a global concern. It is not yet clear how the views and experiences of women and of communities are affected by what they know about the possible risks and benefits of CS. We particularly wanted to know what people say about educational interventions designed to safely reduce CS rates.

# What did we find?

We identified 12 studies, published between 2001 and 2016. Eleven were from high-income countries (USA, Norway, Australia, Canada, UK, Taiwan). One was from a middle-income country (Brazil). Seven involved women who had had a previous CS. We reviewed the studies using qualitative evidence synthesis methods. We found women and communities value educational interventions that include opportunities for dialogue, are individualised (including acknowledgement of previous birth experiences), and are consistent with available clinical care and the advice of health professionals they encounter. We have more confidence in these results for women and communities in high-income countries because of where most of the contributing studies were conducted. Future educational interventions in high income countries should be based on these results, and further research is needed to find out if these interventions are also relevant for middle and low income countries.

## Background

When medically indicated, caesarean section (CS) can prevent deaths and other serious complications in mothers and babies. However, there is evidence of risks for some healthy women and babies undergoing CS [1-3]. The World Health Organization's (WHO) 2015 Statement declares that rates higher than 10% are not associated with reductions in maternal and newborn mortality, and can cause surgical complications, disability or death, particularly in settings that lack the facilities and/or capacity to properly conduct safe surgery [3]. Several countries now have national CS rates above 50%. The average global CS rate is 18.6%, ranging from 6.0 to 27.2% in the least and the more developed regions [4]. In any setting, women with term pregnancies, with a single fetus and a cephalic presentation (Robson Groups 1–5) are the main contributors to rates representing between 75 and 80% of all CSs conducted [5].

Since 2015, global concern about unnecessary CS has increased [3, 6, 7]. Data from 2008 suggest 6.2 million women undergo unnecessary CS annually, at an estimated cost of 2.32 billion US dollars [8]. Despite the global concern and parallel research conducted, effective interventions tested to reduce unnecessary CS have been elusive [9]. The reason for this limited success may lie in the multifactorial nature of the increase and the multiple stakeholders involved. Women, healthcare professionals, systems, culture and society are all key players, whose concerns contribute to the current situation. There is evidence from across settings that women have become more active in seeking CS [10–16]. One systematic review suggests higher preference for CS amongst women with a previous CS, and those living in middle-, versus high-income countries [17]. Non-clinical, educational interventions targeted at women have been proposed and tested to reduce unnecessary CS. These interventions include workshops, booklets, decision-aids [9]. Qualitative evidence of how these interventions are perceived and used is a missing, but essential component to inform the design of future strategies targeting women. The aim of this synthesis was to provide new evidence of what women, communities and publics say about the success or failure of educational interventions targeted at them to reduce unnecessary CS, including barriers and facilitators to intervention use.

# Method

This qualitative evidence synthesis used an interpretive meta-ethnography approach [18]. The funder had no role in the conduct of the research. The protocol was published by PROSPERO [19]. A PRISMA checklist [20] is provided as additional information (Additional file 1: Appendix S0).

# Searches

## Electronic search strategies

Systematic search strategies were developed building on preliminary scoping searches, terms used by existing quantitative reviews of interventions to reduce unnecessary CS, [21–23] guidelines developed by the Cochrane Qualitative Research Methods Group, [24–27] and papers detailing strategies for optimising the identification of qualitative studies in CINAHL, [26] MEDINE [27], EMBASE [28] and PsycINFO [29]. An example search strategy is provided as additional information (Additional file 2: Appendix S1). CINAHL, MEDLINE, PsycINFO, EMBASE, Global Index Medicus, POPLINE, and African Journals Online were searched for eligible studies published between 1st January 1985 and the date of final search (22nd March 2017), to identify studies since the first WHO statement on appropriate technology for childbirth [30]. We imposed no language or geographic restrictions.

## Other sources

As retrieval of qualitative research using databases alone is limited, the reference lists of all the included studies and existing quantitative reviews were back and citation chained [21–23]. In addition, key articles cited by multiple authors (citation pearls) were checked on Google Scholar. The authors of published protocols were also contacted [31, 32]. See Fig. 1 (PRISMA Flow Diagram) for an overview of the study identification, screening eligibility and inclusion process.

# Criteria for inclusion and exclusion of studies Types of study designs

All studies utilising a qualitative research design (e.g. ethnography, phenomenology), or qualitative methods for data collection (e.g. focus group interviews, individual interviews, observation, diaries, oral histories), and which used qualitative methods for data analysis (e.g. thematic analysis, framework approach, grounded theory, thematic network analysis) were eligible for inclusion. Studies using mixed methods designs were also eligible where it was possible to extract findings derived from the qualitative component. Studies in which data were collected using qualitative methods, but which did not perform a qualitative analysis (e.g. if qualitative data were only reported using descriptive statistics), were excluded.

# Types of participants

The population of interest was women in general (defined as any woman of reproductive age; pregnant or non-pregnant) and for pregnant women, those that could be considered low risk (i.e. we excluded studies that included only women with multiple pregnancies, or breeches or where the fetal lie was transverse or oblique) but we included women with previous CS. The Robson 10-Group classification (see Additional file 3: Figure S1) were used as an approximation to illustrate the types of women covered by this synthesis because it is an internationally accepted classification and provides a useful framework for action [33, 34]. Women in Robson Groups 1–5 are the main contributors to high CS rates in any setting and likely the groups where more unnecessary CSs concentrate [5].

## Types of intervention

For the purposes of this synthesis an intervention was 'anything considered by study authors as an intervention additional to usual care undertaken with the aim of reducing unnecessary caesarean section' [19]. This definition



was purposively broad to encompass known and unknown interventions [22]. Interventions targeted at women with a breech presentation (Robson Groups 6 and 7), multiple pregnancy (Robson Group 8), transverse or oblique lie (Robson Group 9) or preterm birth (Robson Group 10) were excluded.

Identified electronic records were collated into one database and duplicates removed. Two synthesis authors (CK, SD) independently assessed each abstract to determine eligibility for inclusion against a priori inclusion and exclusion criteria. The full texts of all papers identified as potentially relevant were retrieved and also independently assessed by CK and SD. The view of the third author APB was sought before agreeing on the final list of included studies.

### Qualitative evidence synthesis

The meta-ethnography [18] approach to qualitative evidence synthesis approach used in this qualitative evidence synthesis comprised of five stages 1) Familiarisation and quality assessment, 2) Data extraction, 3) Coding, 4) Interpretative synthesis, 5) CERQual assessment [35]. For further detail see Additional file 4: Box S1: Qualitative Evidence Synthesis Methodology.

### **Reflexivity statement**

CK is a medical sociologist who held prior beliefs about the complexity and interdependency of social factors driving CS rates informed by primary research with women and health professionals in the UK. APB is a medical officer with over 15 years of experience in maternal and perinatal health research and public health and has witnessed the sense of helplessness and the barriers governments experience when trying to reduce unnecessary CS. SD, is a Professor of Midwifery, her interactions with the data were informed by her experience of the barriers clinical staff encounter on the ground when they try to use their clinical judgement and skills alongside personal values and knowledge of the current evidence base, and the views and choices of childbearing women.

# Results

Our electronic searches yielded 8215 citations. We screened 6671 unique records after duplicate removal. We assessed 181 full-text articles for eligibility and included 12 studies in this qualitative evidence synthesis. The included studies, published between 2001 and 2016, were from seven different countries and mostly from urban areas (Table 1) [36–48]. Samples ranged from 5 to 170 participants, consistent with the qualitative approached used. Eleven studies were from high-income settings (USA, Norway, Australia, Canada, UK, Taiwan). One study was from a middle-income setting (Brazil).

Seven studies exclusively involved women with a previous CS (Robson Group 5), two were targeted at non-pregnant women and communities and three were in Robson Groups 1,2 and 5. Three of the twelve studies were qualitative [37, 39, 41] and sibling studies of trials [49, 50] included in the Cochrane Reviews of non-clinical interventions for reducing unnecessary CS [9, 22]. Quality assessment of studies ranged from A to C. Table 1 lists the quality assessment grades assigned. Seven studies were assessed as having no, few, or only some flaws, with credibility, transferability, dependability, and confirmability unlikely to have been affected (A-B). For each of the 12 studies, the intervention is also described in Table 1. Table 2 is the CERQual [35] summary of qualitative findings table. The development from coding of initial concepts to emergent themes into final themes [18] (with supporting data quotes), and the CERQual [35] summary of evidence profile are provided as (Additional file 5: Table S1; Additional file 6: Table S2). Hereafter, findings are reported under thematic headings with emergent theme sub-headings in bold.

# Theme 1: Mutability of women's and community's beliefs about birth: Ambivalence, empowerment and fear

This theme encapsulates how beliefs about birth vary across populations and over time. Individual women's views about vaginal and caesarean birth are neither stable nor mutually exclusive, as information is continuously and concurrently communicated to them pre-, during, and post-pregnancy.

# Women and communities like learning new information about birth

Participants talked about how interventions had opened up new ways of thinking about birth for them, irrespective of parity, [36–48] with some surprised by the actual number of CSs performed and the risks associated with them. This view was typified by a nulliparous woman in the USA: "*I didn't realize there were so many C-sections*. *I kind of always thought that was like the last option*" ([48]: p. 128). Intervention content also brought issues of risk to the fore as illustrated by this participant in the DiAMOND Trial [50] who initially wanted a VBAC, but when the Decision-Aid (DA) proposed an elective CS, changed her mind: "It [DA] educated me in risks that I didn't know about... it's nice to have been able to do this and learn the things" ([41]: p. 900).

# Pregnancy as a time of uncertainty and transformation of thought about birth

Participants in seven studies [37, 40–42, 44, 45, 48] reported a preference for a particular birth method prior to, or early in pregnancy, then later changed their minds. Three of these studies exclusively involved women with

Table 1 Summar	y of included studies and quality	assessment Country and region	Setting	Number of	Tyne of narticinants	Rohson Grouns	Method	Ouality Assessment
			201113	participants	יואר טי אמיייטאמיינט			
McCants [47, 48]	Prebirth educational brochure.	USA, Americas	Urban	16	Non-pregnant women	Community <sup>a</sup>	In-depth interviews	U
Shorten [46]	Internet-based decision aid about birth choices after previous caesarean developed from an earlier paper-based decision aid.	USA, Americas	Urban	σ	Pregnant women	Ś	Semi-structured Interviews	в
Ramvi [45]	Women who took part in a team Midwifery intervention referred because they requested a caesarean because of fear of birth and actually had a vaginal birth.	Norway, European	Urban	L2	Postnatal women		In-depth interviews	A
Basso [44]	Problematizing educational intervention with groups of pregnant women and partners comprising eight face-to-face sessions.	Brazil, Americas	Urban	51	Pregnant women and partners	1 and 2	Convergent Care study	A
David [43]	Dedicated next birth after caesarean telephone helpline manned by midwives to promote safe and successful vaginal birth in a subsequent pregnancy.	Australia, Western Pacific	Unclear	170	Pregnant women	Ś	Content analysis	U
Milne [42]	Decision board comprising decisional aids that provide a standardized base of written and graphic evidence-based information about vaginal and caesarean birth.	Canada, Americas	Urban	40	Non-pregnant women	Community <sup>a</sup>	Semi-structured interviews	U
Frost [41]	Two computer decision aids (an information programme and an individualised decision analysis programme)	UK, European	Urban	30	Pregnant women	Ś	Semi-structured interviews	в
Farnworth [40]	Informational DVD/video and a home visit by a midwife	UK, European	Urban	18	Pregnant women	5	Semi-structured interview	В
Emmett [39]	Two decision aids (an information programme and an individualised decision analysis programme)	UK, European	Urban	26	Pregnant and postnatal women	L7	Semi-structured interview and observation	В
Wang [38]	Web-based education program for vaginal birth after caesarean	Taiwan, Western Pacific	Urban	10	Postnatal women	5	Telephone interviews	U
Shorten [37]	Birth choices booklet: Development, preliminary evaluation and pilot.	Australia, Western Pacific	Urban	21	Pregnant women	Ŋ	Questionnaire booklet with open questions	U
Cleeton [36]	Suzanne Arms' 1998 videotape Giving Birth	USA, Americas	Sub-Urban	65	Non-pregnant women and male students	Community <sup>a</sup>	Questionnaire with open questions	Ø

<sup>a</sup>Community defined as men and women of reproductive age but not currently pregnant to best of author's knowledge

# Table 2 CERQual summary of findings table

Review finding	Contributing studies	Assessment confidence in	Explanation of confidence in the evidence
Theme 1: Mutability of women's and community's be	iefs about birth. Ambiya	lence Empowerment and Fear	r
Women and communities liked learning new information about birth: The content and design of interventions opened up new ways of thinking about vaginal and caesarean birth for women and communities. Women described how educational interventions informed them about risks and benefits of vaginal birth and caesarean section that were hitherto unknown. Some women were surprised by the actual number of caesareans performed and the risks associated with them. Interventions brought issues of risk to the fore and forced pregnant women in particular to think through more clearly what mattered to them.	[36–47]	High confidence high-income countries and moderate confidence for whole review population	12 studies with minor methodological limitations. Rich data from 7 countries across 3 geographical regions with highest rates of unnecessary caesarean section. No or very minor concerns about coherence.
Women described pregnancy as an inherent time of uncertainty and transformation of thought about birth method: While some women described being very sure about their preferred method of delivery prior to or early in the current pregnancy, many of these same women later changed their minds following experience of an intervention or in response to evolving circumstance. Educational interventions played an important role in helping women prepare for any eventually and to reconcile the benefits of their actual birth method when it did not correspond to their preference.	[37, 40–42, 44, 45, 47]	Moderate confidence	7 studies, 4 of which had very minor methodological limitations. Data from Europe and the Americas only. Very minor concerns about coherence.
Communication of new knowledge and/or support can be empowering: Learning risks were small, or what labour actually entails, enabled some women to feel more prepared and more confident to labour, especially were information about physiological processes was combined with emotional support. Pregnant or recently delivered women described how they had used information to gain control in the pursuit of informed decision- making; either by using the information to inform more meaningful dialogue with health professionals, or to justify a pre-existing preference for either birth method.	[36–42, 44, 45]	Moderate confidence	9 studies with minor methodological limitations. Sufficiently rich data from 7 countries across 3 geographical regions with highest rates of unnecessary caesarean section. No or very minor concerns about coherence.
Some information can provoke fear: Some women and communities found intervention content alarming. Childbirth education video content was described as too gory by a few nulliparous students. Some pregnant women said the use of computer or DVD decision-aids for VBAC increased their anxiety. Use of a decision-aid, combined with follow-up by a midwife helped mediate pregnant women's concerns about risk in one study, while midwives failing to listen to women's concerns and forcing them to birth vaginally compounded fears in another.	[36, 38–41, 45, 47]	Moderate confidence	7 studies with minor methodological limitations. Fairly rich data from USA, UK, Taiwan and Norway. Minor concerns about coherence.
Theme 2: Multiplicity of birth information needs: Fram	ing, format and individu	ual management strategies	
Targeted educational interventions are only one	[36–38, 40–45, 47]	Moderate confidence	10 studies with minor methodological

Targeted educational interventions are only one component informing women's and communities' views and decision-making about birth method: Women describe being exposed to a multiplicity of information sources in their pre-, present- and post-pregnancy trajectories. Some women using decision-aids describe them as "a starting point"; a springboard for seeking more information. Learning from the birth stories of family and friends was widespread. Information was also actively sought in the media and from the Internet, while face-to-face interactions with health professionals were viewed

[36-38, 40-45, 4/]

10 studies with minor methodological limitations. Fairly rich data 7 countries across 3 geographical regions with highest rates of unnecessary caesarean section. No or very minor concerns about coherence with the other 2 studies not attending to this issue.

# Table 2 CERQual summary of findings table (Continued)

	intillaca)		
Review finding	Contributing studies	Assessment confidence in the evidence	Explanation of confidence in the evidence assessment
as the most important influence on actual birth method.			
Desire for educational content conveying the physical work of labour and the social and emotional impact of vaginal and caesarean birth for women. Across settings and education formats, women and communities offered suggestions of what was missing from interventions. They wanted to know more about VBAC and homebirth, what a midwife does, maternity entitlements, the social and emotional impact of caesarean birth, and the "body work" vaginal birth entails. Women also felt vaginal birth could be presented in a more positive way by acknowledging it as an experience. They also wanted information framed in ways women could more easily relate to; for example, many women desired to learn about birth from other women's experiences; some women wanted information about interventions that was personalised.	[36, 37, 39–47]	High confidence high- income countries and moderate confidence for whole review population	11 studies with minor methodological limitations. Data from 6 countries across 3 geographical regions. Very minor concerns about coherence.
Women want multiple modes and formats of educational interventions: Women and communities had wide-ranging views on appropriate language use, figures and tables to communicate information across formats. While many could see the benefits of computer-based interventions, ease of use was problematic for some and pregnant women in particular still desired hard copies of information to revisit and discuss with family members and healthcare professionals. Some concern was expressed about the confidentially of information in on-line decision-aids. Video content was largely welcomed as it facilitated the visualisation of positive, actual birth experiences.	[36–41, 46]	Moderate confidence	7 studies with minor methodological limitations. Data from 3 countries (UK, USA, Taiwan) across three geographical regions. Moderate concerns about adequacy and coherence.
Women desired emotional support alongside the communication of facts and figures about childbirth: Women perceived the choice between vaginal birth and caesarean section as huge; with far-reaching consequences for health and wellbeing. Pregnant women in particular described needing emotional support alongside information about the risks and benefits of birth methods. In tandem with interventions women described additional emotional support from husbands, health professionals and doulas.	[36, 38–41, 43, 45]	Moderate confidence	7 studies with minor methodological limitations. Fairly rich data from 5 countries (UK, USA, Australia, Norway and Taiwan). 6 of the 7 studies involved pregnant or post-natal women faced with the gravity of the actual decision- made.
Theme 3: Interactions with health professionals and in	fluence of healthcare sy	stem: Support, consistency and	d autonomy
Women welcome health professional's acknowledgement of previous birth (or life) experience as an important step in decision-making about future birth method. While previous experiences are important in attitude formation they do not necessarily equate to subsequent preference for delivery method. Across study settings many women and communities valued vaginal birth as natural and a meaningful life experience for women, with fears associated with labour and vaginal birth (pain, uterine rupture) not insurmountable. Few women categorically preferred caesarean section. Some women who had previous experience of caesarean section were particularly keen to avoid it.	[36–38, 40–45, 47]	Moderate confidence	10 studies with minor methodological limitations. Data from 7 countries across 3 geographical regions with richest data from European settings. Minor concerns about coherence.
Intervention content was most useful when it complemented clinical care, was consistent with advice from health professionals and provided a basis for more informed, meaningful dialogue between women and care providers: Some women and communities experiences of interventions suggest they raised more questions than they answered and	[36, 37, 40, 41, 43, 45–47]	Moderate confidence	8 studies with minor methodological limitations. Data from 4 countries across 3 geographical regions. No or very minor concerns about coherence.

Table 2 CERQual summar	of findings	table	(Continued
------------------------	-------------	-------	------------

Review finding	Contributing studies	Assessment confidence in the evidence	Explanation of confidence in the evidence assessment
created a need for additional dialogue with health professionals to discuss issues raised, fears evoked, and revisit birth plans. While some pregnant women described themselves as "desperate" for such conversations, other women were dissatisfied when their expectations went unmet because conversations were too brief, their views were not listened too, the health professional was unknown to them, and/or gave inconsistent advice.			
Women's attitude towards involvement in decision- making: Some women have a strong desire to be involved and to exert control in the decision-making process; others are less certain of their role and value some involvement; while others still are reluctant for any involvement and want qualified health professionals to make the decision for them. The success of any intervention to reduce unnecessary caesarean section is dependent upon pregnant women being open to a role in decision-making and some degree of uncertainty surrounding preference for caesarean section.	[37, 38, 40, 41, 43–47]	Moderate confidence	9 studies with minor methodological limitations. Data from 7 countries across 3 geographical regions. Minor concerns about coherence.
Women are aware of how the organisation of care and information impacts the actual choices available to them: Some women and communities believed intervention content favoured health professionals' hidden agendas to promote whichever method of birth was favoured by them or the hospital or health system in which they work. In two geographical regions pregnant women used metaphors of conflict in the pursuit of their choice of birth method. Other women questioned the exclusion of information about homebirth, excessively high caesarean section rates, and why doctors aren't publically accountable for the number of caesarean sections performed if they are "cutting on women" unnecessarily.	[36–41, 43–45, 47]	Moderate confidence	10 studies with minor methodological limitations. Data from 7 countries across 3 geographical regions. Moderate concerns about coherence.

Page 8 of 14

a previous CS [37, 40, 41]. Two studies reported women who were mildly in favour of a VBAC or who were unsure, changed their preference to an elective CS after the intervention [37, 41]. In one study of non-pregnant women in the community, the change reported was from a preference for CS, to a preference for vaginal birth [42]. Some women talked about the nature of pregnancy as inherently uncertain [40, 41, 44]. Educational interventions had the additional benefit of helping prepare for any eventuality, and to reconcile the benefits of their actual birth method if it did not correspond to their prior preference [41, 44].

# Communication of new knowledge, education and support can be empowering

Some women reported feeling more prepared to labour after intervention exposure, especially where information about physiological processes was combined with support from a midwife or doula [36–42, 44, 45]. Educational content facilitated shifts in confidence, as expressed by a Taiwanese multiparous woman, who "*felt a lot of pain and*"

lost confidence when giving birth last time. I felt very different this time. Because I had taken this course I felt very confident when I was giving birth" ([38]:p5). Similarly, a nulliparous college student in the US commented "this film... turns it [pain] into a form of "power" for women" ([36]:p197). Pregnant, and new mothers also reported how they had used information to gain control; either by using educational content to inform more meaningful dialogue with health professionals, or to justify a pre-existing preference for CS or vaginal birth [40, 41].

# Educational intervention content as anxiety provoking

Childbirth education video content was described as "too graphic" by a few nulliparous students, [36] while some pregnant women described how the use of computer or DVD decision-aid increased their anxiety by communicating "scary information" [40, 41]. Midwifery support helped mediate some pregnant women's concerns, [40, 45] but when midwives and health professionals were perceived as not listening to women's concerns it compounded their fear [45].

This theme highlights the potential of educational interventions to increase anxiety, but also to have positive effects including enhanced knowledge, transformation of beliefs, and confidence to labour, which could reduce CS rates. However, the underlying mechanisms of mutability meant the same interventions were also used in support of elective CS. Reasons for these inconsistencies in effect are explored in theme 2.

# Theme 2: Multiplicity of birth information needs: Framing, format and individual management strategies

This theme encapsulates how recipients of educational interventions differ in terms of levels of health literacy (i.e. familiarity with medical terms), demand for information, and uptake. It also demonstrates how the content is neither delivered, nor received, in a vacuum.

# Educational interventions are only one component informing decision-making about birth method

Women described being exposed to a multiplicity of sources of information [36–38, 40–45, 47]. Women using decision-aids described them as "*a starting point*"; a springboard for seeking more information ([41]:p899). Learning from the birth stories of family and friends was widespread, as was seeking information from the Internet. Face-to-face interactions with health professionals were viewed as the most important influence on actual birth method. Discussions occurring at 36 week antenatal appointments were reported in the UK as limiting the effects of a decision-support intervention delivered earlier in pregnancy ([40]:p121).

# Desire for educational content conveying the physical work of labour and the social and emotional impact of vaginal and caesarean birth

Within and between studies participants identified missing components, including what a midwife does, maternity entitlements, the social and emotional impact of CS, and the physical labour of vaginal birth [36, 37, 39-41, 43, 45–47]. Some participants thought information about vaginal birth should be communicated less as "a medical phenomenon" and more as a personal, spiritual, and emotional experience ([36]:p196). Across interventions, participants wanted information framed in ways they could more easily relate to, particularly watching or reading "responses from [real] people" ([46]:p:394). Interventions with components where obstetric histories and personal characteristics (age, height) could be entered, were welcomed by some, but not all women. If the outcome of the exercise didn't fit with women's prior decisions, they were less likely to find it helpful [41, 46].

# Women want multiple modes and formats of educational interventions

Different women reported different levels of literacy, comprehension or requisite skills and access to electronic resources. While many women reported benefits of computer-based interventions, [38, 39, 41, 46] ease of use was problematic for some [39, 41] as were confidentiality concerns, [46] and an unmet need for hard copies to reflect upon, revisit and share information during discussions with family, friends and health professionals [39, 46]. Varied views on the appropriateness of language, figures, tables, and quizzes to communicate information across formats were reported [36, 38-40, 46]. Some women talked about wanting "not medical terms, kind of straight talking terms and easy facts", found terms like "perinatal unclear", or "out of every thousand women like you patronising" ([39]:p167). Video content was mainly welcomed because it facilitated the visualisation of positive, actual birth experiences, and was easily understood [36, 38, 40].

# Desire for emotional support alongside the communication of facts and figures about birth

Women talked about the decision between vaginal or caesarean birth in the context of their lives [36, 38–43, 45]. Pregnant women in particular described needing emotional support alongside information about the risks and benefits of birth methods. Women with a previous CS specifically needed someone to listen to their previous birth experience, to help them understand what had happened to them and why, to help them go forward in their current pregnancy [40, 43, 45] with midwifery emotional support acknowl-edged by some as a "*turning point*" from preferring another CS to going ahead with a planned vaginal birth [40, 45]. Women valued emotional support from their partners, health professionals and doulas [38–41, 43, 45].

This theme demonstrates the multiplicity of women's birth education needs, the role of individual agency in seeking and managing information, and the importance of how information is communicated. The role of health professional support is explored further in theme 3, alongside the need for consistency between the information imparted in educational interventions and in clinical care, and women's perceptions of who has autonomy over the choice of actual birth method.

# Theme 3: Interactions with health professionals and influence of healthcare system: Support, consistency and autonomy

This final theme reports how shifts in views about birth method are partly shaped by routine interactions with health professionals and health system factors.

# Women welcome health professional's acknowledgement of previous birth (and life) experiences

Women wanted health professionals to acknowledge their prior beliefs and experiences, especially previous traumatic birth experiences, "for the massive thing that it is" to them ([40]:p.120). Previous experiences were important in attitude formation but they did not necessarily equate to subsequent preference for delivery method [36-38, 40-45, 47]. For example, a Norwegian woman (expecting her third child) who requested a CS came to think vaginal birth was possible "on my own terms" ([45]:p271) as a result of the intervention, because she felt her midwife now "knew how she felt" ([45]:p271) and was able to organise care in such a way as to overcome her specific fears. Across study settings many women and communities valued vaginal birth as natural and a meaningful life experience [36-38, 42-44, 47]. While they expressed fears associated with labour and vaginal birth (pain, uterine rupture), they felt these fears could be allayed with appropriate information and support. Few women categorically preferred CS and some with a history of CS were keen to avoid it [37-41, 43].

# Intervention content is most useful when it complements clinical care, is consistent with advice from health professionals and provides a basis for more informed, meaningful dialogue between women and care providers

As reported in Theme 1, interventions could generate more questions than they answered, creating the need for more dialogue with health professionals [36, 37, 40, 41, 43, 45–47]. Multiparous women discussed the tension between receiving fewer opportunities for consultation with health professionals (compared to their first pregnancy), yet feeling there was more to discuss [40, 41]. Some pregnant women described themselves as *"desperate"* for such conversations, with they and others dissatisfied when their expectations went unmet [36, 37, 40, 41, 45–47].

*Women's attitudes towards involvement in decision-making* Some women were highly motivated to be involved in decision-making about birth method, [37, 40, 41, 43, 45] others uncertain of their role or wanted a healthcare professional to make the decision for them [37, 38, 40, 41, 43–47]. Reasons for reluctance included respect for professional knowledge, in contrast to their own, confusion about their right to choose, and readiness to simply see how pregnancy and birth goes.

# Women are aware of how the organisation of care and information can impact the actual choices available to them

Some women and communities were suspicious that intervention content favoured health professionals, hospital or health systems hidden agendas [36–41, 43–45, 47].

Women used metaphors of conflict to describe their perception of the need to engage with health services "armed with information" ([43]:p168) in pursuit of real choice. They anticipated having a "fight on your hands," ([40]:p121) and "dialogue [is] the only weapon" ([44]:p396). Women and men also questioned why doctors are not more publically accountable for "cutting on women" ([48]:p132).

This theme shows recipients of educational interventions understand how health professional preferences and health systems can influence actual birth method. In the final interpretive synthesis stage of the analysis findings were combined to represent our interpretation, through a line of argument.

# Line of argument synthesis

Educational interventions targeted at women are one of multiple factors that influence highly emotive, complex, and fluid decision-making processes during pregnancy and childbirth. In this dynamic, multi-layer context, focused, linear interventions may have limited effects. Prior birth or life experiences, appeared to be important in guiding interpretation of educational materials, including their use to reinforce an existing preference for birth method. They can be empowering and instigate confidence. They also seemed helpful in enabling women to confront the issues where there was ambivalence for birth method. Pregnant and non-pregnant recipients of educational interventions suggest that accurate content is necessary, but not sufficient. Recipients also needed to trust in the information being communicated which may be challenging in the context of certain health systems and alleged hidden agendas. For maternity service users, meaningful interactions with wider social networks, particularly health professionals, had the potential to frame the educational interventions, and to transform expectations and experiences of actual birth method.

# Discussion

### Main findings

An important facilitator of interventions was the widespread appeal of learning about birth. However, this could lead to anxiety as well as empowerment. The acceptability of educational intervention formats and content varied, but a common thread was the importance of communicating information that matters to women. Prior birth experiences, routine interactions (with social networks, health professionals) and continuing information communication (during antenatal appointments, from books or the Internet) of birth information appeared especially important in framing perceptions of intervention value. The potential for the transformation of beliefs about birth, pregnant women's desire for emotional support, and need for dialogue with health professionals acted as facilitators or barriers to positive educational intervention effects. Important barriers to intervention success were inconsistent information content (especially between intervention and routine care) and mistrust in the healthcare system, based on the belief that it would not deliver birth method of choice if this choice did not fit system norms.

## Strengths and limitations

The use of GRADE-CERQual assessment of confidence in findings in this synthesis is a strength [37, 51]. The transferability of the findings across resource settings is limited because eleven of the twelve included studies were from high-income countries where greater emphasis has been placed on women's involvement in decision-making during pregnancy and childbirth. Only one included study was from a middle-income country, Brazil, where rates of CS are particularly high. Furthermore Brazil is considered an upper-middle income country because of the progressive rise of many of its indicators. No studies were from low-income settings, and there were no non-English language publications. The paucity of included studies is an important limitation.

## Interpretation

The present synthesis adds to the emergent evidencebase on reducing unnecessary CS [2-7, 9, 21]. The updated Cochrane review of non-clinical indications to reduce unnecessary CS [9], found only three of the twelve interventions targeted at women or communities had a desirable effect. The qualitative evidence presented in this synthesis suggests women may value emotional experience of care as much as actual birth method per se as an outcome. Women's right to choose their birth method is an important topic in the general CS literature. However, this was not a prominent theme in the current study. Instead women were looking for meaningful and continued interactions with health professionals who could be trusted to provide advice consistent with the educational information received. The review demonstrates that the wider health system (including health professionals) is an important (positive and negative) determinant of women's views and decision-making about birth. This finding resonates with wider literature reporting women's and health professional's views of the reasons behind CS rates [52-54].

The findings also have resonance with existing evidence that women perceive decision-making about birth method as problematic, and that variations exist between what women want to know and what health professionals believe that they should know [55, 56]. Previous studies have suggested that women particularly note a lack information about the benefits of vaginal birth. The absence of evidence on an optimal education format is identified as particularly concerning in quantitative systematic reviews [9]. One systematic review that includes quantitative and qualitative studies suggests the potential of different tools to improve knowledge, and reduce anxiety or decisional-conflict when used at key pregnancy decision points, and in specific circumstances [57]. Women with a previous caesarean (Robson Group 5) could benefit from the more intensive decisional support process these tools facilitate when used as an adjunct to clinical counselling. They can offer women (and clinicians) the opportunity to address their anxieties and the time to consider choice of birth method in partnership.

There has been considerable technological advance during the timespan of the included studies. This synthesis shows how women seek information from multiple sources. The media and communications industry has become a major distributor of information about birth since 2001, with mounting professional concerns about the accuracy and completeness of information [58–60]. A recent meta-synthesis of informal information sources and women's decision-making about birth also suggests the need for a more central role for healthcare providers in the curation and unification of trustworthy information [61]. We identified no eligible studies of celebrity endorsement of birth method, or public dissemination of rates of birth type, but these may prove to be powerful educational tools in the future [16, 62, 63].

Formative qualitative research in local contexts could help improve the design of educational interventions [39, 46, 54] and help build trust, between women, communities and health professionals. It is unclear if the formats discussed here would be accessible in middle and low-income settings, or the availability of providers with time to explain information, and the appropriateness of some content. Future research should focus on evaluating multi-faceted strategies simultaneously targeting women, health professionals, and system change [64–70].

## Conclusion

Women's values and preferences for birth, and for information format and content, vary across populations, and in individual women over time. Unpredictable shifts in views are partly shaped by interactions with health professionals, and by health system factors, as well as specific educational materials. Educational interventions that do not take account of these dynamic factors may have limited effects on the rising CS rate. Tackling unnecessary CS requires educational information with a consistent message, but tailored for the specific needs, values and beliefs of women and communities, alongside emotional support, delivered how and when women need it, in dialogue with health professionals, and reinforced by the health systems encountered by women throughout their pregnancies.

# **Additional files**

Additional file 1: Appendix S0. PRISMA checklist. (DOCX 32 kb)

Additional file 2: Appendix S1. Search strategy. (DOCX 32 kb)

Additional file 3: Figure S1. Robson 10 Group Classification. (JPG 3213 kb)

Additional file 4: Box S1. Summary of qualitative synthesis process. (DOCX 26 kb)

Additional file 5: Table S1. Themes, emergent themes, initial concepts and supporting quotes. (DOCX 35 kb)

Additional file 6: Table S2. CERQual Summary of evidence profile table. (DOCX 33 kb)

### Abbreviations

CERQual: Confidence in the Evidence from Reviews of Qualitative research; CS: Caesarean section

### Acknowledgements

Newton Opiyo for his comments and synergy with the Cochrane Review of Non-clinical interventions for reducing unnecessary caesarean section update. Qian Long and Meghan Bohren for continuing support and discussion about the development of qualitative evidence synthesis and the GRADE-CERQual method for assessing confidence in findings.

### Funding

The funding for this review was provided by the UNDP/UNFPA/UNICEF/ WHO/World Bank Special Programme of Research, Development and Research Training in Human Reproduction (HRP); World Health Organization; and the United States Agency for International Development (USAID).

### Availability of data and materials

All data generated or analysed during this study are included in this published article [and its additional files].

### Author's contributions

APB and CK designed the review with input from SD. CK and SD conducted the searches, identification and screening with agreement by consensus of all authors on final inclusions. CK extracted data, with CK and SD agreeing initial, emergent and final themes. CK, SD and APB all contributed to writing the paper. All authors read and approved the final manuscript.

### Ethics approval and consent to participate

Not applicable

### Consent for publication

Not applicable

### **Competing interests**

"Competing interests: All authors have completed the ICMJE uniform disclosure form at www.icmje.org/coi\_disclosure.pdf and declare: no support from any organisation for the submitted work; no financial relationships with any organisations that might have an interest in the submitted work in the previous three years; no other relationships or activities that could appear to have influenced the submitted work."

# **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

### Author details

<sup>1</sup>School of Community Health and Midwifery, Faculty of Health and Wellbeing, University of Central Lancashire, Preston PR1 2HE, UK. <sup>2</sup>Department of Reproductive Health and Research, World Health Organization (WHO), Geneva, Switzerland.

Received: 14 May 2018 Accepted: 5 July 2018 Published online: 24 July 2018

### References

- Lumbiganon P, Laopaiboon M, Gulmezoglu AM, Souza JP, Taneepanichskul S, Ruyan P, et al. Method of delivery and pregnancy outcomes in Asia: the WHO global survey on maternal and perinatal health 2007–08. Lancet. 2010; 375:490–9.
- Souza JP, Gulmezoglu A, Lumbiganon P, Laopaiboon M, Carroli G, Fawole B, et al. Caesarean section without medical indications is associated with an increased risk of adverse short-term maternal outcomes: the 2004–2008 WHO global survey on maternal and perinatal health. BMC Med. 2010;8:71.
- World Health Organization. WHO statement on caesarean section rates. WHO/RHR/1502; 2015. www.who.int/reproductivehealth/publications/ maternal\_perinatal\_health/cs-statement/en/.
- Betrán AP, Ye J, Moller AB, Zhang J, Gülmezoglu AM, Torloni MR. The increasing trend in caesarean section rates: global, regional and national estimates: 1990-2014. PLoS One. 2016;11:e0148343.
- Vogel JP, Betran AP, Vindevoghel N, Souza JP, Torloni MR, Zhang J, Tuncalp O, Mori R, Morisaki N, Ortiz-Panozo E, et al. Use of the Robson classification to assess caesarean section trends in 21 countries: a secondary analysis of two WHO multicountry surveys. Lancet Global Health. 2015;3:e260–70.
- Miller S, Abalos E, Chamillard M, Ciapponi A, Colaci D, Comandé D, et al. Beyond too little, too late and too much, too soon: a pathway towards evidence-based, respectful maternity care worldwide. Lancet. 2016;388: 2176–92.
- Elshaug AG, Rosenthal MB, Lavis JN, Brownlee S, Schmidt H, Nagpal S, Littlejohns P, Srivastava D, Tunis S, Saini V. Levers for addressing medical underuse and overuse: achieving high-value health care. Lancet. 2017;390: 191–202.
- Gibbons L, Belizán JM, Lauer JA, Betrán AP, Merialdi M, Althabe F. The global numbers and costs of additionally needed and unnecessary caesarean sections performed per year: overuse as a barrier to universal coverage. Working Paper - World Health Report. 2010; http://www.who. int/healthsystems/topics/financing/healthreport/30C-sectioncosts.pdf
- Innie Chen, Newton Opiyo, Emma Tavender, Sameh Mortaz-Hejri, Tamara Rader, Jennifer Petkovic, Sharlini Yogasingam, Monica Taljaard, Sugandha Aggarwal, Jason Wasiak, Pisake Lumbiganon, Russell L Gruen, Suthit Khunpradit, Malinee Laopaiboon, Jeremy Grimshaw, Ana Pilar Betran. Non-clinical interventions for reducing unnecessary caesarean section. Cochrane review Update. In-press.
- Liu X, Landon MB, Cheng W, Chen Y. Cesarean delivery on maternal request in China: what are the risks and benefits? Am J Obstet Gynecol. 2015;212:817.
- Faisal I, Matinnia N, Hejar AR, Khodakarami Z. Why do primigravidae request caesarean section in a normal pregnancy? A qualitative study in Iran. Midwifery. 2014;30:227–33.
- Cook JR, Jarvis S, Knight M, Dhanjal MK. Multiple repeat caesarean section in the UK: incidence and consequences to mother and child. A national, prospective, cohort study. BJOG. 2013;120:85–91.
- Bourgeault DE IV, Sandall J, Wrede A, Vanstone M, van Teijlingen E, De Vries R, Benoit C. Too posh to push? Comparative perspectives on maternal request caesarean sections in Canada, the US, the UK and Finland. Advances in Medical Sociology. 2008;10:99–123.
- Chigbu CO, Ezenyeaku CC, Ezenkwele EP. Obstetricians' attitudes to caesarean delivery on maternal request in Nigeria. J. Obstetrics Gynaecol. 2010; https://doi.org/10.3109/01443615.2010.489165.
- Gama, da Nogueira SG, Viellas, Fernandes E, Schilithz, Corrêa AO, Filha, Theme MM, Carvalho, de Lazaro M, Gomes, Oliveira KR, Costa, Oliveira MC, Leal, do Carmo M. Factors associated with caesarean section among primiparous adolescents in Brazil, 2011-2012. Cadernos de Saúde Pública. 2014; https://doi. org/10.1590/0102-311X00145513.
- Stoll KH, Hauck YL, Downe S, Payne D, Hall WA, International childbirth attitudes- prior to Preganncy (ICAPP) study team. Preference for cesarean section in young nulligravid women in eight OECD countries and implications for reproductive health education. Reprod Health. 2017;14:116.
- Mazzoni A, Althabe F, Liu NH, Bonotti AM, Gibbons L, Sanchez AJ, Belizan JM. Women's preference for caesarean section: a systematic review and metaanalysis of observational studies. BJOG. 2011;118:391–9.
- Noblit GW, Hare RD. Meta-ethnography: synthesizing qualitative studies (qualitative research methods). California: Sage publications, Inc.; 1988.

- Kingdon C, Downe S, Betran A. The use of interventions to reduce unnecessary caesarean sections targeted at women, communities and the public: a qualitative evidence synthesis. PROSPERO 2017 CRD42017059453. Available from: http://www.crd.york.ac.uk/PROSPERO/display\_record.php?ID= CRD42017059453.
- Moher D, Liberati A, Tetzlaff J, Altman DG. The PRISMA group. Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement. PLoS Med. 2009; https://doi.org/10.1371/journal.pmed1000097.
- 21. Chaillet N, Dumont A. Evidence-based strategies for reducing cesarean section rates: a meta-analysis. Birth. 2007;34:53–64.
- Khunpradit S, Tavender E, Lumbiganon P, Laopaiboon M, Wasiak J, Gruen RL. Non-clinical interventions for reducing unnecessary caesarean section. Cochrane Database of Systematic Reviews 2011, Issue 6. Art. No.: CD005528. DOI: https://doi.org/10.1002/14651858.CD.
- Hartmann KE, Andrews JC, Jerome RN, Lewis RM, Likis FE, JN MK, et al. Strategies to Reduce Cesarean Birth in Low-Risk Women [Internet]. Rockville (MD): Agency for Healthcare Research and Quality (US). AHRQ Comparative Effectiveness Review. 2012; Report No.;12(13):EHC128–EF.
- 24. Noyes J, Hannes K, Booth A, Harris J, Harden A, Popay J, Pearson A, Cargo M, and Pantoja T on behalf of the Cochrane qualitative and implementation methods group. Chapter 20: qualitative research and Cochrane reviews. In: Higgins JPT, Green S (editors), Cochrane handbook for systematic reviews of interventions version 5.3.0 (updated October 2015). The Cochrane Collaboration, 2015. Available from http://qim.cochrane.org/supplemental-handbook-guidance
- Booth A. Searching for qualitative research for inclusion in systematic reviews: a structured methodological review. Systematic Reviews. 2016; https://doi.org/10.1186/s13643-016-0249-x.
- Wilczynski NLMS, Hayes RB. Search strategies for identifying qualitative studies in CINAHL Qual Health Res. 2007;17:705–10.
- 27. Wong SSWN, Hayes RB. Developing optimal search strategies for detecting clinically relevant studies in MEDINE. Medinfo. 2004;11:311–6.
- Walters LA, Wilczynski NLMS, Hayes RB. Developing optimal search strategies for retrieving clinically relevant qualitative studies in EMBASE. Qual Health Res. 2006;16:162–8.
- McKibbon KAWN, Hayes RB. Developing optimal search strategies for retrieving qualitative studies in PsycINFO. Eval Health Prof. 2006;29:440–54.
- World Health Organisation. Appropriate technology for birth. In: Lancet; 1985.
   Bermúdez-Tamayo C, Johri M, Perez-Ramos FJ, Maroto-Navarro G, Caño-Aguilar A,
- Garcia-Mochon L, et al. Evaluation of quality improvement for cesarean sections programmes through mixed methods. Implement Sci. 2014;9:182.
- Kaboré C, Ridde V, Kouanda S, Queuille L, Somé PA, Agier I, et al. DECIDE: a cluster randomized controlled trial to reduce non-medically indicated caesareans in Burkina Faso. BMC Pregnancy and Childbirth. 2016;16:322.
- Torloni MR, Betran AP, Souza JP, Widmer M, Allen T, Gulmezoglu M, et al. Classifications for cesarean section: a systematic review. PLoS One. 2011;6:e14566.
- FIGO Working Group on Challenges in Care of Mothers and Infants during Labour and Delivery. Best practice advice on the 10-group classification system for cesarean deliveries. Int J Gynaecol Obstet. 2016;135:232–3.
- Lewin S, Glenton C, Munthe-Kaas H, Carlsen B, Colvin CJ, Gulmezoglu M, Noyes J, Booth A, Garside R, Rashidian A. Using qualitative evidence in decision making for health and social interventions: an approach to assess confidence in findings from qualitative evidence syntheses (GRADE-CERQual). PLoS Med. 2015;12:e1001895.
- 36. Cleeton ER. Attitudes and beliefs about childbirth among college students: results of an educational intervention. Birth. 2001;28:192–201.
- Shorten A, Chamberlain M, Shorten B, Kariminia A. Making choices for childbirth: development and testing of a decision-aid for women who have experienced previous caesarean. Patient Educ Couns. 2004;52:307–13.
- Wang HH, Chung UL, Sung MS. Wu SM. Development of a web-based childbirth education program for vaginal birth after C-section (VBAC) mothers. J Nurs Res. 2006;14:1–8.
- Emmett C, Murphy D, Patel RR, Fahey T, Jones C, Ricketts IW, Gregor P, Macleod M, Montgomery AA. Decision-making about mode of delivery after previous caesarean section: development and piloting of two computerbased decision aids. Health Expect. 2007;10:161–72.
- Farnworth DA, Robson SC, Thomson RG, Burges Watson D, Murtagh MJ. Decision support for women choosing mode of delivery after a previous caesarean section: a developmental study. Patient Educ Couns. 2008;71:116–24.
- Frost J, Shaw A, Montgomery A, Murphy D. Women's views on the use of decision aids for decision making about the method of delivery following a previous caesarean section: qualitative interview study. BJOG. 2009;116:896–905.

- 42. Milne J, Gafni A, Lu D, Wood S, Sauve R, Ross S. Developing and pre-testing a decision board to facilitate informed choice about delivery approach in uncomplicated pregnancy. BMC Pregnancy and Childbirth. 2009;9:50.
- David S, Fenwick J, Bayes S, Martin T. A qualitative analysis of the content of telephone calls made by women to a dedicated 'next birth after caesarean' antenatal clinic. Women and Birth. 2010;23:166–71.
- Basso JF, Monticelli M. Expectations if pregnant women and partners concerning their participation in humanized births. Rev Latino-Am Enfermagem. 2010;18: 390–7.
- Ramvi E, Tangerud M. Experiences of women who have a vaginal birth after requesting a cesarean section due to fear of birth: a biographical, narrative, interpretative study. Nurs Health Sci. 2011;13:269–74.
- 46. Shorten A, Fagerlin A, Illuzzi J, Kennedy HP, Lakehomer H, Pettker CM, Saran A, Wiiterman H, Whittemore R. Developing an internet-decision aid for women choosing between vaginal birth after cesarean and planned repeat cesarean. J Midwifery Womens Health. 2015;60:390–400.
- McCants BM, Greiner JR. Prebirth education and childbirth decision-making. International Journal of Education. 2016;31:24–7.
- McCants BMH. The Impact of Prebirth Education on Childbirth Decision making, 2015. PhD Thesis.
- Shorten A, Shorten B, Keogh J, West S, Morris J. Making choices for childbirth: a randomised controlled trial of a decision-aid for informed birth after cesarean. Birth. 2005;32:252–61.
- Montgomery AA, Emmett CL, Fahey T, Jones C, Ricketts I, Patel RR, Peters TJ, Murphy DJ. DiAMOND study group. Two decision aids for mode of delivery among women with previous caesarean section: randomised controlled trial. BMJ. 2007;334:1305–12.
- Lewin S, Bohren MA, Rashidian A, Munthe-Kaas H, Glenton C, Colvin CJ, Garside R, Noyes J, Tuncalp O, Wainwright M, Flottorp S, Tucker JD, Carlsen B. Applying the GRADE-CERQual approach: making an overall CERQual assessment of confidence and creating a Summary of Qualitative Findings table. Implement Sci 2018;13 Suppl 1:2.
- Béhague DP, Victora CG, Barros FC. Consumer demand for caesarean sections in Brazil: informed decision making, patient choice, or social inequality? A population based birth cohort study linking ethnographic and epidemiological methods. BMJ. 2002; https://doi.org/10.1136/bmj.324.7343.942.
- Habiba M, Kaminski M, Da Frè M, Marsal K, Bleker O, Librero J, et al. Caesarean section on request: a comparison of obstetricians' attitudes in eight European countries. BJOG. 2006;113:647–56.
- Emmett CL, Shaw A, Montgomery AA, Murphy DJ. Women's experience of decision making about mode of delivery after a previous caesarean section: the role of health professionals and information about health risks. BJOG. 2006;113:1438–45.
- Goodhall KE, McVittie C, Magil M. Birth choice following primary caesarean section: mothers' perceptions of the influence of health professionals on decision-making. J Reprod Infant Psychol. 2009; https://doi.org/10.1080/ 02646830801918430.
- Lundgren I, Begley C, Gross MM, et al. 'Groping through the fog': a metasynthesis of women's experiences on VBAC (vaginal birth after caesarean section). BMC Pregnancy Childbirth. 2012; https://doi.org/10. 1186/1471-2393-12-85.
- Dugas M, Shorten A, Dube E, Wassef M, Bujold E, Chaillet N. Decision aid tools to support women's decision making in pregnancy and birth: a systematic review and meta-analysis. Soc Sci Med. 2012;74:1968–78.
- Fioretti B, Reiter M, Betrán AP, Torloni M. Googling caesarean section: a survey on the quality of the information available on the internet. BJOG. 2015;122:731–9.
- Torloni MR, Campos Mansilla B, Merialdi M, Betrán AP. What do popular Spanish women's magazines say about caesarean section? A 21-year survey. BJOG. 2014;121:548–55.
- Whitelaw N. Internet information on birth options after caesarean compared to the RCOG patient information leaflet: a web survey. BMC Pregnancy and Childbirth. 2014;14:361.
- Sanders RA, Crozier K. How do informal information sources influence women's decision-making for birth? A meta-synthesis of qualitative studies. BMC Pregnancy and Childbirth. 2018; https://doi.org/10.1186/ s12884-017-1648-2.
- 62. Kim C, et al. Are league tables controlling epidemic of caesarean sections in South Korea? BJOG. 2005;112:607–11.
- 63. Oleske DM, et al. Information dissemination and the cesarean birth rate: the Illinois experience. Int J Technol Assess Health Care. 1992;8:708–18.11.

- Main IK, Morton CH, Melsop K, Hopkins D, Giuliani G, Gould JB. Creating a public agenda for maternity safety and quality in cesarean delivery. Obstet Gynecol. 2012;120:1194–8.
- Sakala C, Mayberry LJ. Vaginal or cesarean birth? Application of an advocacy organization-driven research translation model. Nurs Res. 2006;55(Suppl):68–74.
- 66. Walsh D, Downe S. Appraising the quality of qualitative research. Midwifery. 2006;22:108–19.
- Downe S, Simpson L, Trafford K. Expert intrapartum maternity care: a metasynthesis. J Adv Nurs. 2007;57:127–40.
- 68. Glenton C, Colvin C, Carl se n B, Swartz A, Lewin S, Noyes J, Rashidian A. Barriers and facilitators to the implementation of lay health worker programmes to improve access to maternal and child health: qualitative evidence synthesis. Cochrane database of systematic reviews 2013, Issue 2. Art. No: CD010414. doi: https://doi.org/10.1002/14651858.CD010414.
- Bohren MA, Vogel JP, Hunter EC, Lutsiv O, Makh SK, Souza JP, et al. The mistreatment of women during childbirth in health facilities globally: a mixed-methods systematic review. PLoS Med. 2015;12:e1001847.
- Qian Long, Ana Pilar Betran, Meghan Bohren, Maria Regina Torloni. Women's and healthcare providers' preference for caesarean section: a mixed methods systematic review. PROSPERO 2016 CRD42016036596.

#### Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

### At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

